CENWP-OP-B                                                                                                                          16 April 2007
MEMORANDUM FOR CENWP-OP, ATTN: Operations Biologist
SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for Week 15 of 2007, which covers the period from 08 through 14 April 2007.
1 OPERATION SUMMARY

1.1 Daily average river flows ranged from 205.2 to 266.9 kcfs.  Daily average project tailwater elevation ranged from 17.3’ to 20.1’ msl.  Daily average powerhouse forebay elevation ranged from 73.2’ to 74.9’ msl.  Daily average river temperature ranged between 47(F and 48(F.  Secchi disk measurements averaged 4.0’ to 6.0’.

1.2 Spill Bays 1 and 18, which supply attraction flow for Cascades Island and B-Branch respectively, are open 0.5’ during daylight hours and are included in the spill pattern.  Daily average spill averaged 1.6 to 101.1 kcfs.  
1.3 Unit Operation. Powerhouse 2 priority.
1.3.1 Unit 8 remains out of service for turbine rehab. 

1.3.2 Unit 10 remains out of service for turbine rehab.
1.3.3 Unit 15 remains out of service due to headgate problems.

1.3.4 Unit 17 went out of service at 1516 on 10 April due to a headgate oil leak.  See 6.1
1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore fish ladder.  The units are placed on standby, as necessary, during nighttime hours to float debris. 

1.5 Gatewell drawdowns are measured on Wednesdays; VBS drawdowns are measured daily on modified units.
2 ACTIVITIES

2.1 AWS Closures.  The following AWS valves were closed for cleaning.
2.1.1 FV1-1 was closed from 0827 until 0901 on 09 April.
2.2 STS/VBS Inspections. No STSs/VBSs were inspected this reporting week.   
Table 1.  STS and VBS inspections.

	Unit
	Date
	Run Hours
	Comments

	11-18
	NA
	NA
	NA


2.3 Dewatering and Fish Salvage.  The following dewatering or fish salvage activities occurred this reporting week.
2.3.1 Unit 10 tail logs were pulled on 09 April.  22 bullheads, 16 sturgeon, two sculpin and one channel catfish were removed and released at the Hamilton Is. boat ramp.
3 RESEARCH
3.1 Battelle.  Monitoring with hydroacoustics near B2CC entrance to study kelt passage.  
3.2 Battelle.  Installed equipment on spillway piernoses for a spillway survival evaluation.
3.3 Fish Field Unit.  Sea lion observations continue.

3.4 NOAA Lamprey.  Construction of the lamprey ramp in the WA shore AWS began in mid January.
3.5 PSMFC Smolt Monitoring.  Three adult steelhead crossed the separator bars.  Spring Creek fish arrived on 13 April.  Increased mortality was seen in the morning.  Please see the 070416 Memo for the Record from Bonneville Fisheries.
3.6 USACE TDG.  New TDG monitors were installed in the spillway tailrace.
3.7 USGS.  Lamprey held in the FERL for tagging evaluation.
3.8 USDA Wildlife Services.  Hazing sea lions from the powerhouses continues. 

3.9 WDFW Fish Counting.  Visual counting continues Bradford Island and Washington Shore
3.10 WDFW/ODFW/CRITFC Sea lion hazing.  Hazing from boats continues.
4 FISHWAYS 

4.1 Project biologists inspected on 08-11, 13-14 April 2007.  
4.2  Fish Passage Plan Observations.
Table 2.  FPP Out of Criteria items.
	DATE
	LOCATion
	FPP itEMS
	CAUSE
	RESPONSE
	TR#

	BRADFORD ISLAND AND BRANCHES

	4/14
	A-Branch weir
	Reading <1.0’>
	FV3-9 not controlling
	Operators adjusted
	

	4/14
	B-Branch weir
	Reading <1.0’>
	See above
	See above
	

	4/12
	Bonneville
	Inspection not done
	Scheduling/FPOM
	New bio hired
	

	Once
	BI and CI
	Calibration not done
	Personnel shortage
	Waiting for new bio
	

	CASCADES ISLAND

	4/13
	FG6-10
	Closed
	Unknown
	Operators opened
	

	WASHINGTON SHORE

	4/8
	Weir 37
	Reading <1.0’>
	In manual
	
	

	4/14
	Weir 38
	Reading <1.0’>
	Unknown
	
	

	All week
	Velocity meter
	Reading <1.5 fps
	Unknown
	Meter TR’d
	


4.2.1 PH1 electronic velocity meter is in an eddy.  Due to this placement, the velocity meter reads less than 1.5 fps.  Readings to meet FPP criteria are taken from the south end mechanical meter.

4.2.2 PH1 PCC diffusers FG2-22A and FG2-19 are closed or partially open.  Repairs scheduled for 07/08.  

4.2.3 FG3-3 stuck partially open.  Repairs are scheduled for winter 07/08.  TR#07-2660

4.2.4 FG3-5 has a damaged shaft and the valve won’t operate.  Repairs are scheduled for winter 07/08.  

4.2.5 FG3-6 stuck partially open.  Repairs are scheduled for winter 07/08.  

4.3 Fishway Temperatures.  Measurements are taken three weirs down from the count station at Bradford Island and at the junction of the UMT and the main ladder at Washington shore. Temperatures are in Fahrenheit.
Table 3.  Bonneville Fishway Temperatures.

	Date
	WS average
	BI average

	4/8
	47.9
	48.0

	4/9
	48.2
	48.1

	4/10
	48.1
	48.0

	4/11
	48.3
	48.3

	4/12
	48.4
	48.2

	4/13
	48.5
	48.5

	4/14
	49.1
	49.0


4.4 Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week.  

4.5 Adult Fishways. 
4.5.1 Calibration is checked on the weekends.  
4.5.2 B-Branch and Cascades Island PLCs are in automatic.
4.5.3 SLEDS installed at all main fish entrances. 

4.5.4 Stab plates remain installed in the FOGs due to the sea lion barriers not fitting properly.
4.6 Juvenile Fishways.
4.6.1 DSM1 continues to remain OOS.
4.6.2 DSM2 orifice lights continue to be lighted with the halogen bulbs.
4.7 Lamprey Fishways.  
4.7.1 The Bradford Island lamprey ramp remains in service as of 23 March.
5 CONSTRUCTION
5.1 B1 ITS Wall Removal.  Plans are to remove the wall, slope the floor, and install three automated gates.
5.2 PH1 STS/VBS removal and TIES removal.  Six TIES and 23PH1 STSs/VBSs will be scrapped.

5.3 PH2 VBS mods.  All units are complete.  Units 14 and 18 are scheduled for winter 07/08.
6 HAZMAT, SPILLS AND CLEANUP –
6.1 Over a ten day period, a total of approximately 314 gallons of hydraulic fluid leaked from a hydraulic Headgate into the Columbia River.  The leak, coming from Headgate slot 17-B has been isolated and repaired.  An investigation in underway to understand the problem and provide recommendations for early detection from a system not designed with the needed protection.  The problem was corroded flange bolts, which allowed the connection to leak.
7 TELETYPES AND OTHER NOTEWORTHY ITEMS
7.1 A teletype sent on 22 December details tailwater operations for chum incubation.   Effective as soon as possible, a hard tailwater constraint to operate project tailwater no lower than 13.0 feet during all hours.  One pulse per day with a minimum duration of 20 minutes and minimum elevation of 13.5 feet. 
7.2 A teletype sent on 06 April requests juvenile spill start at 0001 on 10 April.  Bonneville will spill 100K 24 hours a day.  The spill table may be found in the 1007 FPP page Bon-51, Table Bon-15.  At no time will the project voluntarily spill above the 120% gas cap.

7.3 A teletype sent on 13 April requests that available units be run outside the 1% efficiency range from 0500 until 1700 on 16 April.  This operation will coincide with the scheduled powerhouse transmission line outage scheduled for 0700 until 1700 on 16 April.  This operation will also precede spilling in excess of the gas cap.
7.4 On 13 April, smolt monitoring personnel reported increased mortality in Spring Creek fish.  Please see the 070416 Memo For the Record distributed by Bonneville Project Fisheries.  If you need a copy, please contact Tammy Mackey at 541-374-4552.

7.5 A teletype sent on 13 April requests a change in powerhouse priority and operation to reduce impacts to juvenile fish passing through the juvenile monitoring facility.  Effective immediately, the project is to operate as follows, while spilling 100 kcfs:

7.5.1 Operate Units 14 and 18 within the 1% efficiency

7.5.2 Operate available units (11-13, and 16-17) at the low end of 1%.
7.5.3 Operate available PH1 units within the 1% efficiency.

7.5.4 Spill to the gas cap.

7.5.5 Operate PH1 units outside the 1% efficiency to meet flow requirements.
7.6 A teletype sent on 15 April approves an exceedance of the 24 hour tailwater fluctuation limit of 4’.  This is effective 0001 through 2359 on 16 April.  This tailwater exceedance is needed to reduce spill during the 16 April transmission line outage (see7.3).
JAMES R. MAHAR, P.E.

Operations Manager
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